Author's response to reviews

Title: IP-10 response to RD1 antigens might be a useful biomarker for monitoring tuberculosis therapy

Authors:

Basirudeen Syed Ahamed Kabeer (basir2020@gmail.com)
Alamelu Raja (alamelurja@gmail.com)
Balambal Raman (balambal.r@trcchennai.in)
Satheesh Thangaraj (Satheesh.THANGARAJ@eu.biomerieux.com)
Marc Leportier (marc.leportier@eu.biomerieux.com)
Giuseppe Ippolito (giuseppe.ippolito@inmi.it)
Enrico Girardi (enrico.girardi@inmi.it)
Philippe-Henri Lagrange (philippe-henri.lagrange@sls.aphp.fr)
Delia Goletti (delia.goletti@inmi.it)

Version: 4 Date: 12 May 2011

Author's response to reviews: see over
Journal: BMC Infectious Diseases, MS ID : 1960214574462002.

Title : IP-10 response to RD1 antigens might be a useful biomarker for monitoring tuberculosis therapy

Author list: Basirudeen Syed Ahamed Kabeer, Alamelu Raja, Balambal Raman, Satheesh Thangaraj, Marc Leportier, Giuseppe Ippolito, Enrico Girardi, Philippe-Henri Lagrange and Delia Goletti

Please below find the answers to the questions.

Formatting changes to be made:

-----------------------------

Major revisions (we require the author to make these changes)

Authors' contributions: Please indicate that all authors read and approved the final manuscript. Only 9 of 10 author's are currently mentioned.

Answer: WE ARE 9 AUTHORS.

References: References should be cited in the text using consecutive numbers in square brackets. Reference 45 is mentioned before reference 44 in the text (page 16). Please update the reference list to amend this.

Answer: WE MODIFIED ACCORDINGLY.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Email address: please include email addresses for all the authors in a separate section on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com
Please remove the email addresses from the affiliations section.

Answer: WE MODIFIED ACCORDINGLY.

Highlighting: Please remove all highlighting from the manuscript.

Answer: WE MODIFIED ACCIRDINGLY.

Structure:. Please reformat the manuscript to use the headings "Background" instead of "Introduction", "Methods" instead of "Materials and Methods" and please Insert the heading "Conclusions" above the section of text beginning "In conclusion...".

Answer: WE MODIFIED ACCIRDINGLY.

References: the reference list should contain up to 30 authors¿ names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors (as seen in
references 16, 17, 24, 26, 27, 29, 33, 36 and 44). Please also separate references 38 and 39 in the reference list.

**ANSWER: WE MODIFIED ACCIRDINGLY.**

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

*I have added a reference (new reference 37).*

Kind regards

Delia Goletti, MD, PhD
Clinical Investigator
Laboratorio del Vecchio, Room 13
Head of Translational Research Unit of the Department of Epidemiology and Preclinical Research
National Institute for Infectious Diseases
Via Portuense 292
Roma 00149
Italy
Tel: +39-06-55170-906
Fax: +39-06-5582-825
E-mail address: delia.goletti@inmi.it;
http://www.inmi.it/ricerca_traslazionale.html
http://www.inmi.it/ricerca_traslazionale_english.html